Author's response to reviews

Title: The effect of neonatal vitamin A supplementation on growth in the first year of life among low-birth-weight infants in Guinea-Bissau: two by two factorial randomised controlled trial

Authors:

Sofie Biering-Sørensen (s.biering@bandim.org)
Henrik Ravn (hin@ssi.dk)
Ane B Fisker (a.fisker@bandim.org)
Luis Camala (watoluis@hotmail.com)
Ivan Monteiro (Ivan130978@gmail.com)
Peter Aaby (p.aaby@bandim.org)
Christine Benn (cb@ssi.dk)

Version: 2 Date: 3 December 2012

Author's response to reviews:

Dear Romeo Atienza Jr.

Thank you for reading the paper “The effect of neonatal vitamin A supplementation on growth in the first year of life among low-birth-weight infants in Guinea-Bissau: two by two factorial randomised controlled trial”.

The following changes have been made:

1) In the Methods section on page 6 under the heading setting, we have stated that a parent or guardian of the children invited to participate had to sign or fingerprint a consent form before the children were included in the study.

2) In the abstract we have defined the abbreviation ‘BCG’ at its first use

On behalf of all authors

Sofie Biering-Sørensen